Author's response to reviews

Title: TCF7L2 gene polymorphisms do not predict susceptibility to diabetes in tropical calcific pancreatitis but may interact with SPINK1 and CTSB mutations in predicting diabetes

Authors:

Swapna Mahurkar (mahurkar@ccmb.res.in)
Seema Bhaskar (seema@ccmb.res.in)
Nageshwar D Reddy (aigindia@yahoo.co.in)
Swami Prakash (spami@ccmb.res.in)
Venkat G Rao (aigindia@yahoo.co.in)
Shiv Prasad Singh (sudhasingh@satyam.net.in)
Varghese Thomas (drvt@sancharnet.in)
Giriraj Ratan Chandak (chandakgrc@ccmb.res.in)

Version: 6 Date: 30 July 2008

Author's response to reviews:

Formatting changes as suggested.